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Abstract 

Vanadium redox flow battery (VRFB) is emerging as an efficient and promising technology in 

terms of long-term cycling stability, constant power output, and outstanding safety for large 

scale applications. The crucial component of the VRFB system is the membrane/separator, 

which mitigate the cross mixing of electrolytes yet to provide ionic conductivity; but due to its 

high cost, it limits the commercialization of the VRFB system. The ion exchange membranes 

(IEMs) developed till date as an alternative to the commercial Nafion membrane could not be 

sufficient from the cost effectiveness due to their tedious and complex synthetic processes. A 

flawless membrane should have excellent proton exchange capability with minimum area 

resistance, low swelling ratio and sufficient water uptake along with restricted vanadium ion 

permeability. In this context, low-cost and negligible ionic resistance of the porous separators 

makes it a potential alternative to the high cost Nafion membranes. However, severe vanadium 

ion crossover through the micron size pores of the porous separator needs to be addressed. To 

tackle the same, three membrane/separators were explored and employed in the VRFB 

applications. The research work is focussed on the modification of the existing porous 

membranes/separators.  

Effect of the loading of the Nafion solution is used as a filler in to the porous polyvinylidene 

fluoride (PVDF) membrane for reducing the crossover of the vanadium ion species. Scanning 

electron microscopy (SEM) confirmed the uniform penetration of the Nafion solution 

throughout the membrane. It is was found that water uptake, Nafion consumption (5 wt.%) and 

permeability of vanadium ions of PVDF-Nafion membrane are significantly smaller than the 

Nafion117 membrane. Vanadium permeability of PVDF-Nafion composite membrane reduced 

18 times than that of the Nafion117 membrane. On flip side, the PVDF-Nafion composite 

membrane shows reduced ion conductivity. Therefore, in this work, a strategy is shown how 
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to tackle the additional overpotential due to reduced proton conductivity. Further, the effect of 

the thermal treatment on the porous network of the polyethylene (PE) separator is studied by 

exposing it to the high temperature within the range of 130℃ to 250℃. The vanadium ion 

permeability of the PE-Pristine separator (217×10-7 cm2·min-1) reduced significantly to 9×10-7 

cm2·min-1 for the PE-170 separator owing to the decrease in the pore size. VRFB cell equipped 

with the PE-140 separator demonstrated optimum results in terms of better capacity retention, 

columbic efficiency (CE = 99%) and energy efficiency (EE = 70%). Further, the separator 

performance evaluated at a 3-cell VRFB stack with an effective area increased to 228 cm2. 

Owing to the low EE, the PE separator was further modified wherein a cross-linked polyvinyl 

alcohol (PVA) layer is coated on to the PE separator to mitigate the vanadium ion crossover. 

PE-PVA separator showed better capacity retention with a capacity fade rate of 0.2% Ah·cycle-

1 than the Nafion117 membrane (0.5% Ah·cycle-1). In-situ and ex-situ oxidative stability of the 

separator is explored in detail and a degradation mechanism of the cross-linked PVA membrane 

is proposed accordingly. The synthesized single side coated PE-PVA´ separator exhibits higher 

CE of 99% and comparable EE of 71% at a current density of 120 mA·cm-2 and demonstrated 

excellent lifetime durability tested up to 1600 charge-discharge cycles at 80mA·cm-2. In a 

parallel study, the performance of the recycled PFSA membrane is explored as a potential low-

cost alternative to the Nafion117 membrane recycled from waste membrane. At last, the 

techno-economic assessment is done to justify the practicality of  all the synthesized 

membranes for 1kW 1kWh VRFB system.  

Keywords: Capacity retention; Membrane synthesis; PE separator; Self-discharge; Vanadium 

ion permeability; Vanadium redox flow battery. 
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सार 

वेनेिडयम रेडॉ* +ो बैटरी (VRFB) तकनीक लंबी अविध की साइकिलंग ;<रता, िनरंतर ऊजा? 

उAादन, और बड़े पैमाने पर उपयोग के िलए उGृI सुरKा के मामले मL एक Mभावी और आशाजनक 

तकनीक के Qप मL उभर रही है। VRFB Mणाली का सबसे महUपूण? घटक िझYी/िवभाजक है, जो 

इलेZ[ ोलाइट्स के इंटरिम;*ंग को रोकता है और साथ ही आयिनक प^रवहन Mदान करता है; लेिकन 

इसकी उ_ लागत के कारण, यह VRFB Mणाली के वािण`ीकरण को सीिमत करता है। वािण;`क 

नेिफयन िझYी के िवकb के Qप मL िवकिसत आयन ए*चLज िझ;Yयो ं(IEMs) जिटल और किठन 

संeेषण Mिgयाओ ंके iिIकोण से पया?j नही ंहk। एक बेहतरीन िझYी मL उGृI Mोटॉन ए*चLज Kमता 

होनी चािहए, lूनतम Kेm Mितरोध के साथ, कम सूजन अनुपात, पया?j जल अिधnहण, और Mितबंिधत 

वेनेिडयम आयन पारगoता होनी चािहए। इस संदभ? मL, िछिqत िवभाजक की कम लागत और नगr 

आयिनक Mितरोध इसे उ_ लागत वाली नेिफयन िझ;Yयो ंका िवकb बनाता है। हालांिक, िछिqत 

िवभाजक के माइgोन आकार के िछqो ंसे वेनेिडयम आयन के अिधक gॉसओवर को संबोिधत करने की 

आवsकता है। इसे हल करने के िलए, तीन Mकार की िझ;Yयो/ं िवभाजक का अtयन और VRFB 

अनुMयोगो ंमL उपयोग िकया गया। शोध काय? का मुu उvेs मौजूदा िछिqत िझ;Yयो/ं िवभाजक का 

संशोधन है।  

वेनेिडयम आयन के gॉसओवर को कम करने के िलए िछिqत PVDF िझYी मL िफलर के Qप मL नेिफयन 

सॉxूशन की माmा का Mभाव देखा गया। zैिनंग इलेZ[ ॉन माइgोzोपी (SEM) ने िझYी के भीतर 

नेिफयन सॉxूशन की समान Qप से Mवेश की पुिI की। यह पाया गया िक PVDF-नेिफयन िझYी का 

जल अिधnहण, नेिफयन खपत (5 वज़न Mितशत), और वेनेिडयम gॉसओवर नेिफयन117 िझYी की 

तुलना मL काफी कम है। PVDF-नेिफयन क|ोिजट िझYी की वेनेिडयम पारगoता नेिफयन117 िझYी 

की तुलना मL 18 गुना कम हो गई। दूसरी ओर, PVDF-नेिफयन क|ोिजट िझYी मL आयन प^रवहन मL 
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कमी िदखती है। इसिलए, इस काय? मL यह रणनीित M~ुत की गई है िक कम Mोटॉन प^रवहन के कारण 

उA� अित^रÄ ओवरपोटLिशयल से कैसे िनपटा जाए। इसके अलावा, PE िवभाजक के िछिqत नेटवक?  

पर थम?ल ट[ ीटमLट के Mभाव का अtयन िकया गया, िजसमL इसे 130℃ से 250℃ के उ_ तापमान से 

अवगत कराया। PE-िMÅीन िवभाजक (217×10⁻⁷ सेमी²·िमन⁻¹) की वेनेिडयम आयन पारगoता, िछq 

आकार मL कमी के कारण, PE-170 िवभाजक के िलए काफी घटकर 9×10⁻⁷ सेमी²·िमन⁻¹ हो गई। PE-

140 िवभाजक से लैस वीआरएफबी सेल ने बेहतर Kमता Mितधारण, CE (99%), और EE (70%) के 

संदभ? मL अनुकूल प^रणाम Mदिश?त िकए। आगे, िवभाजक के Mदश?न का मूxांकन 3-सेल वीआरएफबी 

Åैक मL िकया गया, िजसमL Mभावी Kेmफल 228 सेमी² तक बढ़ा िदया गया। कम ऊजा? दKता (EE) को 

tान मL रखते Éए, PE िवभाजक को और संशोिधत िकया गया, िजसमL PE िवभाजक पर gॉस-िलंÑड 

पॉलीिवनाइल अÖोहल (PVA) की परत कोट की गई, तािक वेनेिडयम आयन gॉसओवर को कम िकया 

जा सके। PE-PVA' िवभाजक ने बेहतर Kमता Mितधारण िदखाया, िजसमL नेिफयन117 िझYी (0.5% 

Ah·साइकल⁻¹) की तुलना मL 0.2% Ah·साइकल⁻¹ की Kमता Kय दर पाई गई। िवभाजक की इन-िसटू 

और ए*-िसटू ऑ*ीडेिटव ;<रता का िव~ार से अtयन िकया गया और gॉस-िलंÑड PVA िझYी 

का एक áास तंm M~ािवत िकया गया। संeेिषत िसंगल-साइड कोटेड PE-PVA' िवभाजक ने 120 

mA·सेमी⁻² की धारा घनU पर 99% की उ_ CE और 71% की तुलनीय EE Mदिश?त की, और 80 

mA·सेमी⁻² पर 1600 चाज?-िडàचाज? चgो ंतक परीKण मL उGृI जीवनकाल <ाियU िदखाया। अपिशI 

िझYी से पुनन?वीनीकृत PFSA िझYी के Mदश?न का मूxांकन नेिफयन117 िझYी के िलए संभािवत 

कम लागत वाले िवकb के Qप मL िकया गया। अंततः , 1kW/1kWh VRFB Mणाली के िलए सभी 

संeेिषत िझ;Yयो ंकी äवहाय?ता को सही ठहराने के िलए तकनीकी-आिथ?क मूxांकन िकया गया। 

मु# श%: Kमता Mितधारण; िझYी संeेषण; PE िवभाजक; ã-Mवृिå (सेç-िडàचाज?); वैनािडयम 

आयन पारगoता; वेनेिडयम रेडॉ* +ो बैटरी । 
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